Author's response to reviews

Title: Photographic Protocol for Image Acquisition in Craniofacial Microsomia

Authors:

Carrie L Heike (carrie.heike@seattlechildrens.org)
Laura P Stueckle (laura.stueckle@seattlechildrens.org)
Erik T Stuhaug (erik.stuhaug@seattlechildrens.org)
Luiz A Pimenta (pimental@DENTISTRY.UNC.EDU)
Amelia F Drake (amelia_drake@med.unc.edu)
Daniela Vivaldi (daniela_vivaldi@DENTISTRY.UNC.EDU)
Kathleen CY Sie (kathleen.sie@seattlechildrens.org)
Craig B Birgfeld (craig.birgfeld@seattlechildrens.org)

Version: 3 Date: 23 December 2011

Author's response to reviews: see over
Dear Members of the Editorial Board for Head and Face Medicine,

We are pleased to submit this revised manuscript entitled “Photographic Protocol for Image Acquisition in Craniofacial Microsomia” for consideration of publication in Head and Face Medicine as a methodology article.

We thank the reviewers for their thoughtful feedback. We have made the following changes to the revised submission:

**Major revisions**

- Consent: We updated the consent section to indicate that the participant’s parent signed the consent.
- References: We updated reference 1 and checked the rest of the reference list.

**Minor revisions**

- Header: We removed the header.
- Author list: We formatted the author list as a paragraph.
- Email address: we updated the email addresses for all the authors on the title page, using the following format: AB: abcd@institution.ac.uk
- Abbreviations: We re formatted these in to a paragraph.
- Typography: We checked the manuscript for typographical errors and made final minor corrections and revisions.
- We updated figure 2 to replace the capital “I” from the title with “I” for view C. We also changed the legend for view G to “brows elevated”, and changed the legend for view G in Figure 6 to match.

Thank you for your consideration of our manuscript. Please contact me with any questions.

Sincerely,

Carrie Heike, MD, MS
Children’s Craniofacial Center
Seattle Children’s Hospital
University of Washington
(206) 987-2528